Author's response to reviews

Title: Relationship between Diabetes and grayscale fractal dimensions of retinal vasculature among Indian population

Authors:

Behzad Aliahmad (behzad.aliahmad@rmit.edu.au)
Dinesh K Kumar (dinesh@rmit.edu.au)
Marc G Sarossy (marc@sarossy.com)
Rajeev Jain (drrajeevjain@gmail.com)

Version: 2
Date: 4 July 2014

Author's response to reviews:

Based on the feedback from editorial board, I am submitting a revised version of the manuscript with page and line numbers included.